Author's response to reviews

Title: Parvovirus B19 infection and severe anaemia in Kenyan children: a retrospective case control study

Authors:

James W Wildig (jameswildig@optusnet.com.au)
Yvonne Cossart (ycossart@usyd.edu.au)
Norbert Peshu (npeshu@kemri-wellcome.org)
Nimmo Gicheru (ngicheru@kilifi.kemri-wellcome.org)
James Tuju (jtuju@kilifi.kemri-wellcome.org)
Thomas N Williams (twilliams@kilifi.kemri-wellcome.org)
Charles R Newton (Cnewton@kilifi.kemri-wellcome.org)

Version: 4 Date: 15 March 2010

Author's response to reviews:

Reviewer Giogio Bedogni

Minor essential revision:

Thank you for your comments. I have now recalculated the p values regarding B19 IgM, IgG and PCR among cases vs controls using the exact McNemar's test from StatXact version 8 (see results section and table 3). However, even when using the StatXact software the Odds Ratio for the High IgM 2x2 table (included in table 3) is calculated as +Inf, and the CI 95 for the odds ratio as [1.65, +Inf]. I have included these results as is in the table.